Author's response to reviews

Title: Diffuse duodenal nodular lymphoid hyperplasia: a large cohort of patients etiologically related to Helicobacter pylori infection

Authors:

Mehnaaz S Khuroo Dr (mkhuroo@yahoo.com)
Naira S Khuroo Dr (naira_sultan@yahoo.com)
Mohammad S Khuroo Prof (khuroo@yahoo.com)

Version: 6 Date: 27 March 2011

Author's response to reviews:

BMC Gastroenterology, MS: 5175930324767503, formatting changes
Formatting changes made:
-----------------------------

Major revisions

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match.
Ans: done.

Structure: Please insert a heading titled 'Conclusions' after the discussion.
Ans: done.

Authors' contributions: Please use initials instead of full name.
Ans: done.

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].'
Ans: done.

Minor revisions

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, 'The Title of My Prestigious Publication: A Scientific Study.' should be changed to 'The title of my prestigious publication: a scientific study'.
Ans: done.

Title page: Please display the authors' names, affiliations and email addresses in
separate sections on the title page.
Ans: done.

Text: Please ensure that all text is displayed in black throughout the manuscript.
Ans: done.

Funding: Please move the statement of funding to an Acknowledgements section.
Ans: done.

References: the reference list should contain up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
Ans: done.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements.
Ans: done as desired.

Tables: the table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.
Ans: done for table 1.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Ans: have gone over the document and made changes as needed.